Author's response to reviews

Title: Flow situations during everyday practice in a medical hospital ward. Results from a study based on experience sampling method

Authors:

Åsa Bringsén (asa.bringsen@hkr.se)
Göran Ejlertsson (goran.ejlertsson@hkr.se)
Ingemar H Andersson (ingemar.andersson@hkr.se)

Version: 5 Date: 31 January 2011

Author's response to reviews:

Dear Editorial Production Team

Thank you for your patience and comments. I have now changed the order of the references in the text as well as in the referencelist. Table 5 is now also cited in the text on page 13 (third section). The visible lines in table 3 is now the same size as in the other tables.

Best regards Åsa Bringsén